Author's response to reviews

Title: Quality of life associated with treatment adherence in patients with type 2 diabetes: a cross-sectional study

Authors:

Yolanda V Martínez (yolandamtz@uiessags.com)
Carlos A Prado-Aguilar (carlospa@uiessags.com)
Ramon A Rascon-Pacheco (alberto.rascon@imss.gob.mx)
Jose J Valdivia-Martinez (jjvaldivia@uiessags.com)

Version: 6 Date: 9 July 2008

Author's response to reviews: see over
We did some final changes to the text

On page 1. We deleted the apostrophe from the title.

On page 2. In the methods section, we wrote the next phrase “Quality of life was measured through the WHOQOL-100 (the WHO Quality of Life questionnaire)”, line 8.

On page 9. The mean age was incorrect, the correct one is 58.7, first paragraph, line three.

In the same page. The median HbA1c was incorrect, the correct one is the mean, first paragraph, line six.

In the same page. We changed the information regarding the association between quality of life and adherence in the third paragraph to after the fourth paragraph. This information corresponds to the table 4, the phrased “with the exception of spirituality” was deleted.

In the same page, fourth paragraph, there was two errors regarding the number of the tables, the corrects are table 2 in the line 10 and table 3 in the line 12.

On page 10. The 10 % was incorrect, this was changed for the correct one, second paragraph, line 13.

On page 15. In the acknowledgements section we added a paragraph with information about the YVM thesis.

On page 22. We changed the phrase “** Spearman’s rho” into 5 point font.

On page 23. We separated the information of the level of independence domain from p-value.